### Comments

**Reviewer #1:**

**Specific Comments to Authors:**

A personal description of the role of metaanalysis, and its weaknesses, with regards to butorphanol in epidural labor analgesia. It is well written, reasonable and has appropriate references. It can be published.

(1) The language classification is Grade B. Please visit the following website for the professional English language editing companies that we recommend: [https://www.wjgnet.com/bpg/gerinfo/240](https://www.wjgnet.com/bpg/gerinfo/240).

(2) Self-cited references: There are 3 self-cited references (4-6). The self-referencing rate should be less than 10%. Please keep the reasonable self-citations (i.e. those which are most closely related to the topic of the manuscript) and remove all other improper self-citations. If the authors fail to address the critical issue of self-citation, the editing process of this manuscript will be terminated.

(3) The title cannot start with "the, a, an".

(4) To ensure the accuracy of the references, please use "Edit References by Auto-Analyser" ([https://www.f6publishing.com/Forms/main/ArticleReferenceTool.aspx](https://www.f6publishing.com/Forms/main/ArticleReferenceTool.aspx)) to edit the references of the manuscript.


**Running title:** A short running title of no more than 6 words should be provided. Abbreviations are permitted.